Author's response to reviews

Title: Effectiveness of ophthalmic solution preservatives: a comparison of latanoprost with 0.02% benzalkonium chloride and travoprost with the sofZia preservative system

Authors:

Gerard Ryan Jr (jerry.ryan@pfizer.com)
Joel M Fain (joel.m.fain@pfizer.com)
Cherie Lovelace (cherie.lovelace@pfizer.com)
Karl M Gelotte (karl.m.gelotte@pfizer.com)

Version: 3 Date: 23 March 2011

Author's response to reviews:

Rachel Neilan, MSc
Executive Editor
BMC-series Journals
BioMed Central Floor 6
236 Gray's Inn Road
London, WC1X 8HL

Re: MS: 3212216384477830. Effectiveness of ophthalmic solution preservatives: a comparison of latanoprost with 0.02% benzalkonium chloride and travoprost with the sofZia preservative system Gerard Ryan Jr, Joel M Fain, Cherie Lovelace and Karl M Gelotte

Dear Dr. Cassady-Cain:

We are very pleased with your positive response to our manuscript and much appreciate your noting the need to obtain copyright permission in your communication of March 10, 2011.

Table 1 was developed by the authors from data in the references cited. Thus, this table is not copyright protected. However, Table 2 was reproduced directly from the European Pharmacopoeia. Because this is an official compendium, we overlooked the fact that the actual content is indeed copyright protected and have now received permission from the Directorate for the Quality of Medicines and HealthCare to use the table (permission attached) and have noted such in Table 2 of the revised resubmitted manuscript.

Please contact us if further changes are required.

Sincerely yours,

Gerard Ryan, Jr.
Senior Scientist/Microbiologist
Development Analytics
Pfizer Inc
Eastern Point Road
Groton, Connecticut 06340
Phone: (860) 441-1259
Fax: (860) 715-7738
E-mail: jerry.ryan@pfizer.com